Author's response to reviews

Title: Socioeconomic differences in mortality amenable to health care among Finnish adults 1992-2003: 12 year follow up using individual level linked population register data

Authors:

Alison K McCallum (alison.mccallum@nhslothian.scot.nhs.uk)
Kristiina Manderbacka (kristiina.manderbacka@thl.fi)
Martti Arffman (martti.arffman@thl.fi)
Alastair H Leyland (a.leyland@spsu.mrc.ac.uk)
Ilmo Keskimaki (ilmo.keskimaki@thl.fi)

Version: 2 Date: 29 May 2012

Author's response to reviews:

A separate methods section was added to the abstract as requested.